Author's response to reviews

Title: Adult-onset leukoencephalopathy, brain calcifications and cysts: a case report

Authors:

Gang Cheng (yjscg2003@126.com)
Yaming Wang (yamingwang@sohu.com)
Chao Dong (flyingperson85@126.com)
Jianning Zhang (jnzhang2005@gmai.com)
Yuhong Meng (584056134@qq.com)

Version: 2 Date: 3 December 2012

Author's response to reviews:

I made revisions in the first paragraph of the discussion (red mark) and explained how the case report significantly advance on what we already know from the published literature about LCC. There was the first case report about LCC with visual field change.